Author's response to reviews

Title: Early and very early hepatocellular carcinoma: when and how much do staging and choice of treatment really matter? A multi-center study.

Authors:

Fabio Farinati (fabio.farinati@unipd.it)
Adriana Sergio (adry_80@libero.it)
Anna Baldan (anna_baldan@virgilio.it)
Anna Giacomin (annagiacomin@katamail.com)
Maria Anna Di Nolfo (annadinolfo@tin.it)
Paolo Del Poggio (pdpoggio@ospedale.treviglio.bg.it)
Luisa Benvegnu (luisa.benvegnu@unipd.it)
Gianludovico Rapaccini (rapaccini@rm.unicatt.it)
Marco Zoli (marco.zoli@unibo.it)
Franco Borzio (frborzio@tin.it)
Edoardo G Giannini (egiannini@unige.it)
Eugenio Caturelli (e.caturelli@tiscalinet.it)
Franco Trevisani (franco.trevisani@unibo.it)

Version: 4 Date: 23 January 2009

Author's response to reviews: see over
To: Joseph Dunckley  
Assistant Editor  
BMC-series journals  

Object: third revised version of manuscript MS: 3762728082196621  

Padua, 23 January 2009  

Dear Mr Dunkley,  

Enclosed you will find the re-revised version of the above paper. As requested, we introduced the formatting changes requested.  

Major revisions  

1. We modified the Author list, omitting the name of the ITA.LI.CA. group from title page and mentioning it only in the Acknowledgements.  
2. We modified the Tables, avoiding vertical lines.  

Minor revisions  

3. Figures: please note that we tried to reduce the white space around figure 1 without success; we hope it will be possible for you to modify the figure, as mentioned in the letter we received.  

With many thanks for your collaboration.  

Prof. F.Farinati